Author's response to reviews

Title: Efficacy of depression treatments targeting first-generation immigrants: Results from a systematic review

Authors:

Josefine Antoniades (Josefine.antoniades@monash.edu)
Danielle Mazza (Danielle.mazza@monash.edu)
Bianca Brijnath (bianca.brijnath@monash.edu)

Version: 2 Date: 20 March 2014

Author's response to reviews:

Dear Editor and Team,
Thank you for alerting the research team to the issues in our paper “Efficacy of depression treatments targeting first-generation immigrants: Results from a systematic review”.

We have amended the following as requested:

1. Added Acknowledgement statement:
"The authors would like to acknowledge the support of the funding bodies which have facilitated this study: JA was funded to undertake this research by an Australian Postgraduate Award. BB is financially supported by a fellowship from the Australian National Health and Medical Research Council. None of the authors declare any conflict of interest."

2. As our review did not require ethical approval, we have added the following statement in the conclusion: "This study did not require ethical approval”.

3. Tables placed after the Reference section.

4. In addition to the changes made as per the editorial team’s request, I noted that a figure legend was missing from the document, so I have added the Figure legend “Figure 1. PRISMA flowchart of literature search process” after the tables.

Thank you again for your prompt feedback. I hope the changes are satisfactory, if not we are of course happy to make further amendments.

With Thanks,
Josefine Antoniades